Minutes & action items July 20, 2005 TRT Barrel integration.

· Fuse box:

· No fuse failure in new fuse box at Duke ( 4.5 weeks now. 

· In PNPI 1st batch of fuses was mounted in fuse boxes on 13/07/05. 17 out of 408 were lost due to learning, 570 were mounted correctly without losses.
· 2nd batch of 1000 fuses is available. The fuse boxes brought to CERN by Jack were hand-carried to PNPI by Yuri Ryabov. They are in Leonid's lab since Monday this week. No firm promise when the next 1000 fuses will be mounted, as they also still have to assemble WEBs in parallel (hope it will be done end of this week or early next week).
· It was suggested to have a set of production fuse box completed for use in system test as early as possible. Harold will discuss with Fido for the possibility. 
· We also want to set up a production line in 154 for assembling HV cables and connectors. Jack was working with Herve on that before he left.   
· Sector test:
· Mechanical part of sector test: cooling, active gas, HV for the top quadrant is tested. Electronics part of sector test is proceeding and focusing on the sector for system test. 
· It looks like we can separate mechanical and electronics part of the sector test. The mechanical part is much simpler and can be done quickly after manifold is mounted. Hopefully, we can at least finish mechanical part of sector test by the end of Aug.  
· Manifold:

· The next set of manifold is supposed to ship near end of this week. It’s not clear whether it will be all of the rest or only the 3rd quadrant (the bottom one). The manufacturer is having some problem getting mounting ears from their sub-contractor. Situation will be clearer in a week or so. 

· System test: 

· Fuse box for system test is having current draw problem on the filter capacitors and are being worked on. ( although there was one bad Capacitor that shorted, the main effect of current draw was traced to the cooling of the front face of the module when the electronics power was turned off- and cooling still running). 

· Most electronics components are in hand except filter board. Expect to have both ends of stack 8 and 7 powered by Monday. HV will be attached by that time, too. 

· Front harness on 1 stack was connected and being tested. Data looks encouraging. Data taken with HV on does not show any substantial difference, but more testing is planned.

· Daq, display, data base (including dcs), analysis software are being worked on. Plan to have a demonstration next Wednesday.  

· Cosmic ray telescope should be worked on Thursday by students and Anatoli.

